Author's response to reviews


Authors:

Milagros Rico-Blazquez (milagros.rico@salud.madrid.org)
Esperanza Escortell-Mayor (eescortell@salud.madrid.org)
Isabel del-Cura-Gonzalez (isabel.cura@salud.madrid.org)
Teresa Sanz-Cuesta (teresa.sanzcu@salud.madrid.org)
Pilar Gallego-Berciano (mariadelpilar.gallego@salud.madrid.org)
Gonzalo delas-Casas-Cámara (lalocascam@yahoo.es)
Sonia Soto-Díaza (sonia.soto@salud.madrid.org)
Petra García-Sanz (pgarciasanz@salud.madrid.org)
Natalie Harris-De-La-Vega (natalie.harris@salud.madrid.org)
María Martin-Martin (maria.martin.martin@salud.madrid.org)
Lorena Dominguez-Perez (ldominguezp@salud.madrid.org)
Araceli Rivera-Alvarez (araceli.rivera@salud.madrid.org)
Juan-Antonio Sarrion-Bravo (jantonio.sarrionb@salud.madrid.org)
Concepcion Perez-De-Hita (mconcepcion.perez@salud.madrid.org)
Marisol De-Frias-Redondo (mariasoledad.frias@salud.madrid.org)
Carmen Ferrer-Arnedo (cferrer@salud.madrid.org)
Montserrat Hernandez-Pascual (montse.hernandezpa@salud.madrid.org)
Antonio Valdivia-Perez (antonio.valdivia@marinasalud.es)

Version: 3 Date: 7 October 2013

Author's response to reviews: see over
To whom it may concern:

We are glad to send you the manuscript titled ‘CUIDACARE project: effectiveness of a nursing care intervention on the quality of life of elders in a caregiver role. A cluster-randomized clinical trial’, for you to valuate the possibility of including it as a Study Protocol article in BMC Nursing.

This study is being developed by Gerencia de Atención Primaria of Madrid and it has been supported by a research grant from Spanish Ministry of Economy.

To assure the quality of the manuscript we respect the Consort statement. This work has not been sent to any other scientific journal yet.

This study was funded by PN of I+D+I 2008-2011 and the ISCIII – Subdirección General de Evaluación y Fomento de la Investigación (PI11/02132).

This is a review in this journal, in response to your email of October 4, forwarding protocol version with the following changes suggested:

1. - Title page: we have include a title page at the front of the manuscript file. It contain the names, institutions, countries and email addresses of all authors, and the full address of the Submitting post author. We have included 5 new affiliation of the authors.

2.- The Spanish Science and Innovation Ministry and Agency Pedro Lain Entralgo have evaluated the protocol as a precursor to the granting of aid for research. Enclosed in this mail files that support the concessions. However, these agencies do not send the investigator documents establishing the review. Only information attached grant aid.

If you have any questions, please feel free to contact us in the following e-mail address.

Yours sincerely,

Milagros Rico-Blázquez
Esperanza Escortell-Mayor
Isabel del-Cura-González
Teresa Sanz-Cuesta
Pilar Gallego-Berciano
Gonzalo de las Casas-Cámara
Sonia Soto-Díaz
Petra García-Sanz
Natalie Harris-de-la-Vega